Author's response to reviews

Title: PET-CT in the sub-arctic region of Norway 2010-2013. At the edge of what is possible?

Authors:

Jan Norum (jan.norum@unn.no)
Ursula Søndergaard (ursula.sondergaard@unn.no)
Erik Traasdahl (erik.traasdahl@unn.no)
Carsten Nieder (carsten.nieder@nlsh.no)
Geir Tollåli (geir.tollali@helse-nord.no)
Gry Andersen (gry.andersen@unn.no)
Rune Sundset (rune.sundset@helse-nord.no)

Version: 2 Date: 11 August 2014

Author's response to reviews: see over
You will find enclosed an article titled: PET-CT in the sub-arctic region of Norway 2010-2013. The article deals with the four years experience running a PET-CT scanner in the Norwegian sub-arctic region and documents patients examined, distances travelled and availability for patients in the region.

The paper includes now page numbers.

We think the paper may be of interest to health care owners who plan to install a PET-CT scanner in a remote area. This as the paper shows what is possible at the “edge of the Arctic”.

Best regards

Jan Norum MD/PhD
Head of Department of Radiology